Author's response to reviews

Title: Rapidly progressive Bronchiolitis Obliterans Organising Pneumonia presenting with pneumothorax, persistent air leak, acute respiratory distress syndrome and multi-organ dysfunction: a case report

Authors:

Indranil Chakravorty (i.chakravorty@herts.ac.uk)
William LG Oldfield (William.Oldfield@St-Marys.nhs.uk)
Carlos MH Gomez (Carlos.Gomez@St-Marys.nhs.uk)

Version: 10 Date: 5 April 2008

Author's response to reviews:

The header line has been removed and the affiliations have been completed in full according to the expected format. The abstract as been updated as in the manuscript.